Answering Reviewers

Dear editor at World Journal of Methodology,

We greatly appreciate the review of our manuscript. We have considered the reviewer comments, revised our manuscript and are now submitting the revised version.

The detailed response to each point raised by the reviewer is described below.

Reviewer 1 (02861117):

Thank you for the comments about this review. We have made some grammatical changes throughout the manuscript, as asked. Changes in the body of the text and addition of words and punctuations. In addition, we added the suggested information in the “Laboratory findings” topic, based on the proposed articles.